Author’s response to reviews

Title: Medical student experiences in prison health services and Social Cognitive Career Choice: a qualitative study

Authors:

Ron Brooker (r.brooker@westernsydney.edu.au)
Wendy Hu (w.hu@westernsydney.edu.au)
Jennifer Reath (j.reath@westernsydney.edu.au)
Penelope Abbott (p.abbott@westernsydney.edu.au)

Version: 3 Date: 13 Dec 2017

Author’s response to reviews:

Dear Editor,

We have addressed all the revisions requested as follows

1) List of Abbreviations

Please provide a list of all the abbreviations used in the manuscript. This list should be placed just before the Declarations section. All abbreviations should still be defined in the text at first use. - completed

2) Consent to Publish

Please clarify within this section whether you received written informed consent from participants to publish anonymous quotations. - completed

3) Author Contributions

Please include the initials and individual contributions of every author within this section. - completed

4) Clean Manuscript
At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files. - completed.